Author’s response to reviews

Title: Co-existence of Phenylketonuria and Fabry Disease on a 3 year-old boy: Case report

Authors:

Daniela Concolino (dconcolino@unicz.it)
Maria Rapsomaniki (maria.rapso@virgilio.it)
Eliana Disabella (e.arbustini@smatteo.pv.it)
Simona Sestito (simona@galaxia.it)
Maria G Pascale (giusypascale@virgilio.it)
Maria T Moricca (dconcolino@unicz.it)
Giuseppe Bonapace (giusebona@libero.it)
Eloisa Arbustini (e.arbustini@smatteo.pv.it)
Pietro Strisciuglio (pietro.strisciuglio@unina.it)

Version: 3 Date: 26 March 2010

Author’s response to reviews: see over
Dear Editor,

thank you for your help regarding our manuscript “Co-existence of Phenylketonuria and Fabry Disease on a 3 year-old boy: Case report”. We modified the paper according to your suggestions as follows:

Major revisions:
Author list: we written the list of authors exactly as they are in the submission system

Minor revision.
Title page: we removed the word Title preceding the main heading
Figures: we minimised white space around the image
Typography: we checked and corrected a typography errors

We hope that the revised paper is now suitable for the publication as short report.

I look to hearing soon from you

Best wishes

Pietro Strisciuglio M.D.
Professor of Pediatrics